Author's response to reviews

Title: Depression and anxiety, an Indicated Prevention (DIP) protocol in homes for the elderly: feasibility and (cost) effectiveness of a stepped care programme

Authors:

Els E Dozeman (els.dozeman@vumc.nl)
Anneke DJF van Schaik (a.van.schaik@ggzba.nl)
Aartjan ATF Beekman (aartjanb@ggzba.nl)
Wim WAB Stalman (w.stalman@vumc.nl)
Judith JE Bosmans (judith.bosmans@falw.vu.nl)
Harm HWJ van Marwijk (hwj.vanmarwijk@vumc.nl)

Version: 5 Date: 8 March 2007

Author's response to reviews:

A section "Figure legend" is included